

African schistosomes in small mammal communities: Perspectives from a spatio-temporal survey in the vicinity of Lake Guiers, Senegal

Julien Kincaid-Smith, Boris A. E. Savassi Sègnito, Bruno Senghor, Christophe Diagne, Youssoupha Niang, Mamadou Kane, Caroline Tatard, Carine Brouat, Laurent Granjon

▶ To cite this version:

Julien Kincaid-Smith, Boris A. E. Savassi Sègnito, Bruno Senghor, Christophe Diagne, Youssoupha Niang, et al.. African schistosomes in small mammal communities: Perspectives from a spatio-temporal survey in the vicinity of Lake Guiers, Senegal. PLoS Neglected Tropical Diseases, 2024, 18 (12), pp.e0012721. 10.1371/journal.pntd.0012721. hal-04868523

HAL Id: hal-04868523 https://hal.inrae.fr/hal-04868523v1

Submitted on 6 Jan 2025

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Distributed under a Creative Commons Attribution - NonCommercial 4.0 International License



G OPEN ACCESS

Citation: Kincaid-Smith J, Savassi BSAE, Senghor B, Diagne C, Niang Y, Kane M, et al. (2024) African schistosomes in small mammal communities: Perspectives from a spatio-temporal survey in the vicinity of Lake Guiers, Senegal. PLoS Negl Trop Dis 18(12): e0012721. https://doi.org/10.1371/ journal.pntd.0012721

Editor: Monique Léchenne, Swiss Tropical and Public Health Institute: Schweizerisches Tropenund Public Health-Institut, SWITZERLAND

Received: May 31, 2024

Accepted: November 25, 2024

Published: December 23, 2024

Copyright: © 2024 Kincaid-Smith et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The cox1 and ITS sequences from individual samples are available in S3 File and were deposited in GenBank (NCBI) under accession numbers PP658465 - PP658508 (cox1) and PP658638 - PP658972 (ITS).

Funding: The author(s) received no specific funding for this work.

Competing interests: The authors have declared that no competing interests exist.

RESEARCH ARTICLE

African schistosomes in small mammal communities: Perspectives from a spatiotemporal survey in the vicinity of Lake Guiers, Senegal

Julien Kincaid-Smith¹*, Boris Sègnito A. E. Savassi^{2,3}, Bruno Senghor⁴, Christophe Diagne¹, Youssoupha Niang⁵, Mamadou Kane⁵, Caroline Tatard¹, Carine Brouat¹, Laurent Granjon¹

1 CBGP, IRD, CIRAD, INRAE, Institut Agro, Université de Montpellier, Montpellier, France, 2 Centre de Recherche pour la lutte contre les Maladies Infectieuses Tropicales (CReMIT/TIDRC), Université d'Abomey-Calavi, Abomey-Calavi, Bénin, 3 IHPE, Univ Montpellier, CNRS, IFREMER, Univ Perpignan Via Domitia, Perpignan, France, 4 VITROME, Campus International IRD-UCAD de Hann, Dakar, Sénégal, 5 CBGP-BIOPASS 2, IRD, Campus IRD-ISRA de Bel-Air, Dakar, Sénégal

* kincaid-smith.julien@ird.fr

Abstract

Schistosomiasis is a neglected tropical disease of public health significance. In view of its elimination as a public health problem by 2030, adopting a One Health approach is necessary, considering its multidimensional nature. Animal reservoirs, in particular, pose a significant threat to schistosomiasis control in Africa and beyond. In this study, we conducted a spatio-temporal survey of Schistosoma infections in small mammal communities and intermediate snail hosts in the vicinity of Lake Guiers in northern Senegal. Sampling campaigns were undertaken four times between April 2021 and August 2022 around eight villages. A total of 534 small mammals of four species, primarily Hubert's multimammate mice Mastomys huberti, were captured. Out of 498 individuals examined, only 18 rodents (17 M. huberti and 1 Arvicanthis niloticus) were infected with schistosomes. The infection rates in M. huberti varied over time (prevalence range: 2.4% to 9.3%, intensity range: 4 to 132), and space (prevalence range: 3.1% to 40%, intensity range: 2 to 110) and were higher in adult hosts captured during or just after the rainy season, a time when older individuals dominate in rodent populations. Using a multi-locus molecular approach (cox1 and ITS) on Schistosoma larvae (cercariae and miracidia) and adult worms, we identified Schistosoma mansoni as the most widespread species. We also detected Schistosoma bovis and Schistosoma haematobium in M. huberti from one locality (Temeye). Although no Schistosoma hybrids were found, the discovery of a male S. mansoni and a female S. bovis pair raises concerns about potential hybridization patterns that could occur in rodents. Finally, three snail species were found infected (25 Biomphalaria pfeifferi, 3 Bulinus truncatus and 1 Bulinus senegalensis) including with S. mansoni, S. bovis, S. haematobium and S. haematobium x S. bovis hybrids. Our findings highlight the spatial-temporal variations of Schistosoma infections in rodents and emphasize the need for fine-scale monitoring over time and

space for effective One Health measures and ensuring the sustainability of schistosomiasis control efforts.

Author summary

Schistosomiasis is a neglected tropical disease of major human and veterinary health concern. To ensure its sustainable control, the role of animal reservoirs that could maintain and spread the disease needs a better understanding. We investigated the spatio-temporal variations of Schistosoma infections in small mammal communities around Lake Guiers, northern Senegal. Small mammal species diversity was relatively low at transmission sites. Most infected rodents were found during or just after the rainy season, a time when older individuals dominate in rodent populations. From a spatial perspective, transmission was highly focalized and infection parameters (prevalence, intensity) highly variable, making it challenging to establish clear patterns. Schistosoma mansoni, responsible for human intestinal schistosomiasis, was the most widespread species found. However, sporadic infections with species of human (S. haematobium) and animal significance (S. bovis), including a case of mixed infections (S. mansoni x S. bovis) in rodents raises concerns about hybridization events that may occur in these hosts. Our findings confirm the role of rodents as reservoir or spill-over hosts for Schistosoma in northern Senegal and suggest that fine scale monitoring of animals at transmission sites could help to ensure the sustainability of schistosomiasis control.

Introduction

Schistosomiasis is a neglected tropical disease (NTD) of major global health concern. According to the World Health Organization (WHO) over 250 million people are infected and approximately 200,000 deaths are reported annually [1,2]. The disease is endemic to Africa, Asia, the Middle East, South America and the West Indies, however the highest burden is recorded within sub-Saharan Africa [3]. The causative agents of the disease are digenean trematodes of the genus Schistosoma. In Africa, the major species of public health concern are Schistosoma mansoni and Schistosoma haematobium, causing intestinal and urogenital schistosomiasis, respectively [4]. Schistosomes have a complex life cycle involving specific freshwater snails as intermediate hosts (i.e., planorbs of the genus Biomphalaria for S. mansoni and Bulinus for S. haematobium) and a wide range of definitive vertebrate hosts, including human and other animals [5]. Infection occurs when the definitive host comes in contact with contaminated freshwater during daily activities. Larvae of the parasites, known as cercariae, released from their specific molluscs, actively penetrate the host skin and develop into adult worms. The pairing of male and female worms leads to the production of hundreds of eggs daily, which are excreted in the definitive host's urine (S. haematobium infection) or faeces (S. *mansoni* infection), perpetuating the parasite's life cycle. Some eggs remain trapped in the host's tissues and are responsible for the pathology [4].

Despite substantial efforts to control schistosomiasis and the effectiveness of mass drug administration (MDA) using praziquantel (PZQ) in reducing global infections [6], the disease persists in endemic areas and has even re-emerged in previously controlled regions [7]. The establishment of schistosomiasis outside its endemic range, in Europe (Corsica, France), since 2013 [8,9] further highlights the need for comprehensive control strategies. In order to achieve

the ambitious goals set by the WHO to eliminate schistosomiasis as a public health problem and ultimately complete interruption of transmission in selected regions by 2030 [3,10], a One Health approach is necessary [11,12]. This approach takes into account the multidimensional nature of this disease including the role of animal reservoirs and the potential multi-host, zoonotic transmission of schistosomes. Previous examples from the zoonotic species *Schistosoma japonicum* in Asia (infecting over 40 species) [13–16], but also *S. mansoni* in rodents in South America (Brazil) and the West Indies (Guadeloupe) [17–20] have illustrated the threat animal reservoirs and multi-host dynamics constitute to the sustainability of schistosomiasis control strategies. However, in Africa, our understanding of the host specificity of human- and animal-infective *Schistosoma* spp., the epidemiology of multi-host parasites and the role of animal reservoirs in human infection and disease control remains incomplete.

While *S. mansoni* is primarily associated to human infections, it has been found in over 50 different vertebrate hosts (see [21] for review) including a majority of non-human primates [22–25] and small mammals, particularly rodents [17–20,26–31]. Although their contribution to the transmission is poorly documented throughout Africa, small mammals have been found naturally infected with *S. mansoni* in several countries including Senegal [26–30], Ethiopia [32] and Kenya [31,33]. The main host species for *S. mansoni* are rodents of the Muridae family (*Mastomys huberti* and *Arvicantis niloticus* in Senegal, *Mastomys natalensis, Lophuromys flavopunctatus, Pelomys isseli* and *Aethomys kaiseri* in Kenya, whilst host species were not reported in the study from Ethiopia) and in rare cases shrews (*Crocidura olivieri* in Kenya).

Concerning S. haematobium, this species was until recently considered as a strictly humanspecific parasite [34,35]. It has been found only on very rare occasions in other hosts [5,36,37], including rodents, with no evidence of a significant role in transmission. However, the epidemiological situation of urogenital schistosomiasis has become more complex in recent years, in particular due to the discovery of *Schistosoma* hybrids [38]. Indeed, humans have been diagnosed with urogenital schistosomiasis caused by hybrids between S. haematobium and S. bovis (a typical parasite of ruminants [26,27,31,39,40]) in several African countries [38,41–50], and following an outbreak in southern Europe [9,51,52], suggesting that host-specificity breakdown has already occurred, and further raising concerns on the zoonotic nature of African schistosomes. Moreover, the finding of a West African hybrid strain in Europe has alerted on the adaptive and invasive capabilities of these pathogens, likely to further complicate control strategies [9,51]. Recent studies showed that S. haematobium x S. bovis hybrids could also infect cattle in Benin [49] as well as rodents in Benin [53], Senegal [26], and in Corsica [54], thus supporting their zoonotic transmission on a local scale. However, compared to Benin where rodents and cattle seem to actively participate to transmission, only few rodents with low infection intensities and no cattle were found harbouring *Schistosoma* hybrids in Senegal [26] or Corsica [54]. This may suggest that the parasite multi-host transmission patterns may be dynamic and/or specific through space and potentially time [27].

To date, the global picture suggests that small mammals are suitable reservoirs for both intestinal and urogenital schistosomiasis caused by hybrids. Rodents are of particular interest for several reasons. First, schistosomes have a strong affinity for rodents, which are commonly used in laboratory settings to maintain various *Schistosoma* species [55]. Second, their capacity to live as commensals in close contact with humans make them good candidates as main reservoirs of zoonotic diseases [56,57]. Third, their large population sizes, wide distributions and high reproductive capacities, further contribute to their potential as key hosts in the transmission of zoonotic pathogens [12,58]. Last, rodents are unique in that they are the only known hosts to carry a diversity of *Schistosoma* spp. that infect both humans (*S. mansoni*, *S. haematobium* x *S. bovis* hybrids) and animals (*S. bovis* and *S. haematobium* x *S. bovis* hybrids). Rodents, claimed as "biotic hubs" [26,31], may provide a

favourable environment for genetic interactions and/or hybridization of *Schistosoma* species, resulting in the emergence of new strains or hybrids with varying morbidity, host range and drug susceptibility [27,38,59]. All together this highlights the importance of understanding the role of rodents in the global burden of schistosomiasis for effective control strategies.

In this study we conducted a spatio-temporal survey of small mammal communities on the shores of Lake Guiers in Senegal, a endemic region for human [45,47,60,61] and animal [26,27,29,45] schistosomes. Our aim was to describe potential variations in infection levels with *Schistosoma* species across space and time and establish links with host species and individual characteristics, which are important parameters to help adapting the management of the disease.

Material and methods

Ethical statement

The "Centre de Biologie pour la Gestion des Populations" (CBGP) carries out research activities on rodent-parasite interactions and their consequences for health. The establishment holds an approval on using animals for scientific purposes (F34-169-001, prefectural decree n° 24 XIX-042). The project obtained a favourable advice of the Comité Consultatif Éthique pour la Recherche en Partenariat (CCERP) of the Institut de Recherche pour le Développement (IRD) (dated 29 November 2021). Trapping campaigns were organized after informing the village chiefs of the area about the objectives of the project, and obtaining explicit consent from local authorities and land owners. All targeted small mammals are classified as "Least Concern" (LC) by the International Union for Conservation of Nature (IUCN) Red List. The examined animals were treated in accordance with published guidelines on animal welfare and the use of wildlife in research [62]. Captures of non-target species were rare and specimens caught unintentionally were released at their point of capture. In the absence of a national legislation of Senegal for the implementation of the Nagoya Protocol on Access and Benefit-Sharing (ABS), we obtained an exemption from authorization of access and use of genetic resources (n° 001105) issued by the Competent National Authority (the National Focal point on ABS).

Hosts sampling through space and time

Fieldwork was carried out in the vicinity of Lake Guiers in northwestern Senegal (Fig 1A). The climate is of Sahelian type, characterized by an 8-9 month-long dry season and a short rainy season between July and October. Fieldwork was conducted over two years at four occasions, namely in April and November 2021, and in February-March and August 2022, the latter corresponding to the rainy season. For the purpose of this study and because of differences in land usage and access to freshwater sources, two geographical zones separated by roughly 50km were selected. Zone A located north-east of the lake (Fig 1B) is mainly characterized by irrigated crop fields and a large network of irrigation canals serving as potential schistosomiasis transmission sites. Zone B located south of the lake (Fig 1B) relies on traditional cultivation and livestock breeding with schistosomiasis transmission sites suspected to be on the shores of the lake itself. These two zones were also selected because of previous data indicating the presence of *Schistosoma* spp. and/or their hybrids in rodent populations [26,27]. Two main villages were targeted in zone A (1: Temeye and 2: Mbane), and six villages in zone B (3: Gankette Balla, 4: Guéo, 5: Gankette Guent, 6: Keur Momar Sarr, 7: Merina Guewel, and 8: Diokhoul) (Fig 1B). Within or adjacent to these villages, sampling sites corresponding to potential transmission foci were defined (based on knowledge of the frequentation of these sites by the local populations) where small mammals were captured (Fig 1C). The exact position with GPS coordinates of each trapping site is available in <u>S1 File</u>.



Fig 1. Map of the study area showing locations sampled during the survey. A) Geographical situation of the Lake Guiers in northwestern Senegal. **B)** Localization of zones A and B and main villages sampled around the lake: 1. Temeye, 2. Mbane, 3. Gankette Balla, 4. Guéo, 5. Gankette Guent, 6. Keur Momar Sarr, 7. Merina Guewel, and 8. Diokhoul. **C)** Sampling sites within or adjacent to each village including five sites for Temeye (1a: Canal Sall, 1b: Lewa, Ic: Lewa-Soubalo, 1d: Soubalo, 1e: North), five sites for Mbane (2a: South, 2b: Center, 2c: North_A, 2d: North_B, 2e: North_C), a unique site for Gankette Balla (3), Guéo (4) and Gankette Guent (5), two sites for Keur Momar Sarr (6a: KMS_South, 6b: Feto), four sites for Merina Guewel (7a: Merina Guewel (7b: Merina Guewel 7c: Takh, 7d: Ringaye) and a unique site for Diokhoul (8). The figure was produced using images from Sentinel-2 (January 2022, theia-land.fr) and Natural Earth (naturalearthdata.com) under Qgis (http://www.ngis.org). The base layer of the map (i.e., the country border shape) was downloaded from https://www.naturalearthdata.com/downloads/50m-cultural-vectors/. It contains information from OpenStreetMap and OpenStreetMap Foundation, which is made available under the Open Database License.

https://doi.org/10.1371/journal.pntd.0012721.g001

Trapping of small mammals

Small mammals were captured using locally made wire-mesh live traps (8.5 x 8.5 x 26.5 cm). Trapping sites correspond to various access points to water (including holes in the riparian vegetation composed of *Typha* sp. and *Phragmites* sp.) serving for laundry, bathing and fishing activities as well as irrigation canals and livestock watering sites. The traps were baited with local peanut butter and placed at the end of the afternoon at 5–10 meter intervals for 1 to 3 nights per site and per sampling period. The traps were systematically inspected and removed the following morning to avoid the capture of non-target species and any untimely removal in these frequented sites. Live small mammals were returned to the laboratory, kept in a quiet, shaded environment until they were euthanized for necropsy (within four hours to minimize the time they spend captive). Information on the date, location, habitat and daily results obtained at each trapping site were stored in the "Small Mammals" database (https://doi.org/10.15454/WWNUPO) hosted by the CBGP.

Post-mortem examinations

Small mammals were euthanized by intraperitoneal injection of sodium thiopental (300 mg/kg body weight) and death was confirmed after cervical dislocation and the absence of pedal reflex. Individuals were identified to the species level based on morphological, ecological and biogeographical knowledge [63]. For each specimen we recorded body weight, gender, reproductive status and external measurements (head and body, tail, hindfoot, and ear lengths) according to standard procedures [63]. Age class (adult/juvenile) was inferred from weight distributions, body length and reproductive status [63]. To account for potential biases in the capture rate (CR) through space and time, a "disturbance index" corresponding to the ratio between half the number of "inactive" traps (i.e., traps closed without any capture, traps found missing, or traps occupied by individuals of non-target species) and the number of trap nights was calculated [64]. Because this index showed low variation across space and over time (see results) the CR was used as a proxy of the relative abundance of each species. It was calculated per night for each trapping site as the number of animals captured divided by the total number of set traps (i.e., the trapping effort) [63].

Diagnosis of Schistosoma infections in small mammals

The presence and abundance of adult worms were used to determine the prevalence and intensity of *Schistosoma* spp. infection in small mammals, respectively. We used the hepatic perfusion technique [65] (slightly adapted to fieldwork conditions) and dissections to diagnose and extract the worms from small mammals blood circulating system. Briefly, a 50 ml syringe was used to perfuse the euthanized animals with a physiological solution of sodium citrate (7.5‰) and sodium chloride (8.5%). The perfusate injected in the left ventricle was collected from the hepatic portal vein and filtered through a polyamide cloth. If any, Schistosoma adult worms were counted and pairs separated under a binocular magnifying glass. Specific organs of each host examined at necropsy (liver, mesenteries and large intestine) were carefully inspected to identify any remaining worms even if the hepatic perfusion was negative. Schistosomes were individually stored at room temperature in annotated 2 ml Eppendorf tubes containing 95% ethanol during fieldwork, then at 4°C upon arrival at the laboratory. Parasite eggs collected from the granulomatous livers of hosts with signs of infection (presence of worms, granulomatous formations due to eggs) were tested for miracidia hatching. Briefly, the liver and the large intestine where Schistosoma eggs accumulate were grounded through a colander and abundantly washed with NaCl solution (8.5%). The suspension obtained was filtered through a series of metal sieves of decreasing mesh size (425, 180, 106 and 45 µm) to concentrate

potential eggs. The final sieve (45 μ m) was rinsed over a Petri dish with bottled spring water at room temperature to recover the eggs. Miracidia hatching from eggs was stimulated in freshwater under indirect sunlight or a lamp [66]. After 20 minutes and up to 6h, the Petri dishes were inspected for the presence of miracidia under a binocular magnifying glass. Larvae were individually collected in 2 μ L using a micropipette before being rinsed in a drop of bottled spring water and transferred onto a Whatman Indicating FTA Classic card (GE Healthcare Life Sciences) for DNA storage [67,68].

Intermediate hosts sampling and parasitological diagnosis

Schistosome intermediate host snails were collected following an opportunistic, non-standardized sampling procedure within or adjacent to some of the small mammal capture sites. We favoured open freshwater sources with abundant vegetation mostly composed of *Ceratophyllum* sp. as favourable habitats for the snails [69]. Snails were collected using a scoop net and species were identified using shell morphology [70]. Only potential hosts for schistosomes were kept (*Bulinus* spp. and *Biomphalaria pfeifferi*). Infection status and associated prevalence in snails were determined after inducing the emission of *Schistosoma* cercariae. Each snail was abundantly washed with freshwater to avoid parasite cross-contaminations and individually transferred to a tube containing bottled spring water. Snails where exposed to indirect sunlight or artificial light over the day to take into account differences in chronobiological rhythms of different *Schistosoma* species so that patent snails release cercariae [67]. The cercariae from positive snails were individually pipetted in 2µL under a binocular magnifying glass onto Whatman Indicating FTA Classic cards for DNA storage [67,68]. Positive molluscs were stored individually while remaining, non-patent molluscs were pooled and stored in 95% ethanol.

Molecular characterisation of the parasites

DNA from adult worms kept in ethanol was extracted using the DNeasy Blood and Tissue Kit (QIAGEN) following the manufacturer's instructions. DNA from larvae (miracidia from rodents and cercariae from snails) preserved on Whatman Indicating FTA Classic cards was processed as described previously [71]. The samples (miracidia, cercariae and adult worms) were first characterized by a rapid diagnostic multiplex polymerase chain reaction (RD-PCR) targeting a partial fragment of the mitochondrial cytochrome c oxidase subunit 1 (cox1) [72,73]. Briefly, the universal forward primer Asmit1 (5'-TTTTTTGGTCATCCTGAGGTG TAT-3') and three species-specific reverse primers, namely ShR (5'-TGATAATCAATGACC CTGCAATAA-3') for S. haematobium (amplicon size: 543 bp); SbR (5'-CACAGGATCAGA CAAACGAGTACC-3') for S. bovis (amplicon size: 306 bp) and SmR (5'-TGCAGATAAAGC CACCCCTGTG-3') for S. mansoni (amplicon size: 375 bp) were used in this study. Reactions were performed in 12.5 μ L containing 2 μ L of DNA template, 1× of Dream Taq green buffer (including 2 mM of MgCl₂), 0.2 mM of dNTP, 0.4 µM of each primer, 1 unit of Dream Taq (Thermo Fisher Scientific, France) and ultra-pure water. Cycling parameters for *cox1* RD-PCR consisted of an initial denaturation at 95°C for 5 min, followed by 40 cycles at 95°C for 30 s, 58°C for 30 s, and 72°C for 1 min, with a final 7 min extension at 72°C. Because of potential hybridization between Schistosoma species, DNA was also amplified and sequenced for the complete (981 bp) nuclear ribosomal DNA internal transcribed spacer (ITS). We used the forward ETTS1 (5'-TGCTTAAGTTCAGCGGGT-3') and the reverse ETTS2 primers (5'-AACA AGGTTTCCGTAGGTGAA-3') [74] and the same PCR reactions as described above. PCR cycling parameters for the ITS gene differed from the cox1 PCR only by increasing the elongation step to 1 min 30 sec. Four microliter of each PCR products were run on a 1.5% agarose gel

for 45 min at 120V to confirm successful amplification and verify amplicon size used to discriminate each species (*cox1*). The ITS PCR products and a subset of randomly chosen *cox1* amplicons (to confirm RD-PCR profiles) were sent to Eurofins MWG (Germany) for purification and sequencing using original PCR primers.

DNA (cox1 and ITS) sequencing analysis

Successful sequences were assembled and manually edited using Sequencher v.4.5 (http:// genecodes.com) to remove ambiguities and sequencing errors. The cleaned sequences were aligned with MUSCLE on MEGA version 7.0.26 [75,76]. Species assignment was performed by comparison with sequences available in GenBank. The identity of the sequence (species and gene) was confirmed using the basic local alignment search tool (BLAST). As hybrids of first generation generally exhibit both parental nuclear rDNA ITS copies, appearing as double peaks on chromatograms, known polymorphic positions between species observed on the ITS gene were verified by visualisation of the original sequence chromatograms to identify heterozygosity [77]. Therefore, hybrids are detected either when the ITS marker shows heterozygous pattern at specific mutation points and/or if ITS and *cox1* markers are discordant in species assignation. The *cox1* and ITS sequences from individual samples are available in S3 File and were deposited in GenBank (NCBI) under accession numbers PP658465—PP658508 (*cox1*) and PP658638—PP658972 (ITS).

Statistical analyses

Generalized linear mixed models (GLMMs) were carried out to evaluate spatio-temporal variations in rodent populations. GLMMs are particularly useful because they allowed us to account for both fixed effects, such as the influence of time and location, and random effects, such as variability between sampling periods or villages. This approach provides a more flexible and robust framework for analysing complex ecological data with repeated measures and hierarchical structure. Three separate models were applied to trap-level data, incorporating both temporal (i.e., 'sampling period') and spatial (i.e., 'zone', 'village') predictors. The models aimed to analyze, as dependent variables: (i) the rodent capture rate (i.e., whether at least one rodent was captured or not; model #1), (ii) the rodent sex (i.e., whether the individual is a female or a male; model #2), and (iii) the rodent age class (i.e., whether the individual is a juvenile or an adult; model #3). Using a three-model approach is essential because each dependent variable—capture presence, sex, and age class—represents a distinct biological aspect of the population and requires separate statistical treatment. This allowed us to address different ecological questions, tailored to the unique structure of each response variable. For each model, we considered a binomial distribution and a model selection approach was performed, using the Akaike information criterion with correction for samples of finite size (AICc) as a goodness-of-fit indicator [78-80]. All predictors were included in the starting models, and the "sampling site" was added as a random factor. Models with all possible combinations of the terms included in the starting model were generated and those with a $\Delta AICc < 2$ with respect to the model with the lowest AICc were selected. Among them, the most parsimonious was chosen based on the fewest number of predictors kept. Deletion testing and log-likelihood ratio (LRT) tests determined the significance of explanatory variables. The assumptions of each final model were checked graphically, by an analysis of their residuals. When needed, post-hoc comparisons were carried out with pairwise Wilcoxon rank sum tests (PWT, 95% family-wise confidence level). We carried out the whole procedure at both interspecific and intraspecific levels (for dominant species, see below) for model #1, but only at intraspecific level for model #2 and model #3. The summary of the most parsimonious GLMMs finally

selected is presented in S1 Table. *Schistosoma* prevalence (i.e., percentage of infected hosts within the sampled host population) were compared over time (sampling periods), across space (zones, villages), and following host-related factors ('age class' and 'gender') using a Pearson's χ^2 test or a Fisher Exact test when sample size expected frequencies was less than 5. The intensity of infection (i.e., number of worms per infected host) in rodents were compared according to the same variables using (*i*) either a Kruskal-Wallis (KW) test for variables with more than two modalities (sampling periods, villages), followed by PWT in case KW tests were significant, or (*ii*) a Wilcoxon—Mann Whitney test for variables with only two modalities (zone, gender). Statistical analyses were performed using DHARMa [81], MuMIn v1.43.6 [82], and lme4 v1.1–8 [83] R packages under R version 4.4.0 (http://www.rproject.org) and Bios-taTGV (http://biostatgv.sentiweb.fr). For all tests, statistical differences were considered significant when p < 0.05.

Results

Small mammal trapping through time and space

A total of 534 small mammals were captured over the course of 4122 trap nights, representing an overall CR of 13.0% over two years. Noticeably, 710 of the 4122 overall trap nights (i.e., >17%) were considered as inactive as they were found either closed without any capture (n = 530), missing (n = 42), or occupied by individuals of non-target species (n = 138). The majority of by-catch individuals were unidentified anuran amphibians (n = 83), followed by birds (*Amaurornis flavirostra* and *Gallinula chloropus*, n = 48) and fishes (catfish and tilapia, n = 7) that were immediately released (S1 File). Our disturbance index showed low variation over sampling periods (range: 7.8% in April 2021 to 10.2% in November 2021) and across villages (range: 7.5% in Mbane to 10.8% in Keur Momar Sarr) thus no correction was applied to the CR calculation based on the total trapping effort (S1 File).

Overall, the targeted rodent species caught comprised 356 *M. huberti*, 137 *A. niloticus*, and 2 *Taterillus pygargus*. Additionally, 39 *Crocidura viaria* shrews were trapped. Species-related CR was in favour of *M. huberti*, which was clearly dominant along the partially flooded shores of the lake and along irrigation canals, often overgrown with reeds, followed by *A. niloticus*, which was well represented in crop fields and gardens adjacent to water bodies, then *C. viaria* and *T. pygargus* (Table 1). Considering all species, model #1 revealed that the overall CR significantly varied over sampling periods from April 2021 to August 2022 (LRT = 28.01, df = 3, p = 3.6e-06). This change in CR over time was significant for *M. huberti* (LRT = 16.53, df = 3, p = 0.0009) in February-March and August 2022 compared to both April and November 2021.

Table 1. Numbers	of trap nigh	hts and individuals	captured for each s	pecies at each sa	mpling	period, with the c	apture rate in brackets.
raole it riamoero	or thep may		aptured for each o	period at each ou		period, min the e	ap care race in oraeneco

Sampling periods	April 2021	November 2021	February-March 2022	August 2022	Total
Nb. trap nights	835	1107	1222	958	4122
Nb. closed-empty	109	167	150	104	530
Nb. missing traps	8	9	10	15	42
By-catch	14	49	28	47	138
Disturbance index	7.8%	10.2%	7.7%	8.7%	8.6%
Mastomys huberti	91 (10.9%)	121 (10.9%)	90 (7.4%)	54 (5.6%)	356 (8.7%)
Arvicanthis niloticus	60 (7.2%)	22 (2.0%)	37 (3.0%)	18 (1.9%)	137 (3.3%)
Crocidura viaria	8 (1.0%)	17 (1.5%)	10 (0.8%)	4 (0.4%)	39 (0.9%)
Taterillus pygargus	0	1 (0.1%)	1 (0.1%)	0	2 (0.5%)
Total captures	159 (19.0%)	161 (14.5%)	138 (11.3%)	76 (7.9%)	534 (13.0%)

https://doi.org/10.1371/journal.pntd.0012721.t001

Zone-Village	Trap nights	Mastomys huberti	Arvicanthis niloticus	Crocidura viaria	Taterillus pyg <mark>arg</mark> us	Total
A-Temeye (1)	973	48 (4.9%)	19 (2.0%)	16 (1.6%)	2 (0.2%)	85 (8.7%)
A-Mbane (2)	640	31 (4.8%)	89 (13.9%)	23 (3.6%)	0	143 (22.3%)
B-GKB (3)	200	12 (6.0%)	10 (5.0%)	0	0	22 (11.0%)
B-Guéo (4)	709	39 (5.5%)	0	0	0	39 (5.5%)
B-GKG (5)	592	75 (12.7%)	2 (0.3%)	0	0	77 (13.0%)
B-KMS (6)	355	67 (18.9%)	3 (0.8%)	0	0	70 (19.7%)
B-MGW (7)	503	74 (14.7%)	12 (2.4%)	0	0	86 (17.1%)
B-Diokhoul (8)	150	10 (10.5%)	2 (2.1%)	0	0	12 (8.0%)

Table 2. Summary of the number of individuals captured through space. Capture rates are indicated in brackets. The code in parentheses for each site refers to Fig 1.

https://doi.org/10.1371/journal.pntd.0012721.t002

The CR significantly varied for *A. niloticus* (LRT = 18.24, df = 3, p = 0.01) only between April 2021 and the other sampling periods (PWT, p < 0.05). Noticeably, after April 2021, crop fields and gardens adjacent to water bodies were less intensively sampled because of the low probability of finding infected individuals there: this may explain the lower CR for *A. niloticus* observed after the first sampling period.

At the spatial level, we did not find significant differences in the overall CR between the geographic zones. However, model #1 suggested significant variations of overall CR between villages within and across geographic zones (LRT = 18.24, df = 7, p = 0.011) with overall values ranging from 5.5% in Guéo to 22.3% in Mbane (Table 2). The number of species captured varied according to the geographical zones. Four species were captured in zone A and only *M*. *huberti* and *A*. *niloticus* trapped in zone B (Table 2). Variations of CR (model#1) between villages were significant at the intraspecific level for *M*. *huberti* (LRT = 25.38, df = 7, p = 0.0007) and *A*. *niloticus* (LRT = 20.88, df = 7, p = 0.004). Nonetheless, variations of CR seemed to suggest opposite trends between *M*. *huberti* and *A*. *niloticus*, the former being overall more abundant in zone B, while the latter being more abundant in zone A. It is however worthy to note that the highest capture rate for *A*. *niloticus* was found in only one locality of zone A where several sites adjacent to crops fields and gardens where sampled (Mbane), *M*. *huberti* being the dominant species in all other localities (Table 2 and S1 File).

Small mammal age and sex population structure

As both *C. varia* and *T. pygargus* were represented by low numbers of individuals subsequent analyses of population sex- (model#2) and age- (model#3) structure focused only on *M. huberti* and *A. niloticus*. We unambiguously attributed 355 and 339 *M. huberti*, and 137 and 128 *A. niloticus* to an age and sex class, respectively. The deficit in the number of individuals attributed to a particular sex in comparison to those attributed to an age class arises from the capture and immediate release of very young individuals which did not have time to become infected with schistosomes. Although their sex was not recorded, these individuals were classified as juveniles. The proportion of juveniles within each species significantly varied over time (*M. huberti*: LRT = 13.03, df = 3, p = 0.005; *A. niloticus*: LRT = 18.55, df = 3, p = 0.0003) with an overall lower proportion of juveniles in August 2022 compared to both April 2021 and February-March 2022 for *M. huberti* (PWT, p-values < 0.05) and in April and November 2021 compared to both February-March or August 2022 for *A. niloticus* (PWT, p-values < 0.05).

At the spatial scale, no significant difference was observed between the overall proportion of juveniles between zone A and B, either for *M. huberti* or for *A. niloticus*, given this variable was not kept in the most parsimonious model selected (Table 3). Furthermore, our GLMMs did not reveal any significant variation of the sex ratio between sampling periods, villages or

Zone	Species	Number of juveniles / total number of individuals captured								
		April 2021	November 2021	February-March 2022	August 2022	Total				
А	Mastomys huberti	2/17 (11.8%)	2/19 (10.5%)	6/24 (25.0%)	0/18 (0%)	10/78 (12.8%)				
	Arvicanthis niloticus	5 /50 (10.0%)	0/15 (0%)	11/28 (39.3%)	7/15 (46.7%)	23/108 (21.3%)				
В	Mastomys huberti	17/74 (23.0%)	16/102 (15.7%)	14/65 (21.5%)	2/36 (5.6%)	49/277 (17.7%)				
	Arvicanthis niloticus	0/10 (0%)	1/7 (14.3%)	2/9 (22.2%)	0/3 (0%)	3/29 (10.3%)				
A & B	Mastomys huberti	19/91 (20.9%)	18/121 (14.9%)	20/89 (22.5%)	2/54 (3.7%)	59/355 (16.6%)				
	Arvicanthis niloticus	5/60 (8.3%)	1/22 (4.5%)	13/37 (35.1%)	7/18 (38.9%)	26/137 (19.0%)				

Table 3. Population age-structure of M. huberti and A. niloticus populations over time and across space. The proportion of juveniles is indicated in parentheses.

https://doi.org/10.1371/journal.pntd.0012721.t003

geographic zones, neither in *M. huberti* nor in *A. niloticus*. The summary of the GLMMs information is presented in <u>S1 Table</u>.

Schistosoma spp. infection in small mammal communities

Amongst the 534 specimens trapped, 498 were examined for Schistosoma infection including 336 M. huberti, 126 A. niloticus, 35 C. viaria and 1 T. pygargus. In total, Schistosoma trematodes were only detected in 18 rodents, representing an overall prevalence of 3.6% (Table 4 and S2 File). No Crocidura shrews or Taterillus gerbils examined were found infected with schistosomes, and a single A. niloticus individual (adult female) was found to be infected (prevalence: 0.8%). Infected *M. huberti* (prevalence: 5.1%) were exclusively adult individuals caught close to freshwater sources (<5 meters). There was no significant difference of prevalence between females (10/154) and males (7/182) *M. huberti* ($\chi^2 = 1.21$, df = 1, p = 0.27). Overall, a total of 191 adult worms were recovered from rodents across all hosts, sampling sites and sampling periods (median: 4, range: 1–52 among carrier hosts): 186 in *M. huberti* (median: 2, range: 1-52) and 5 in A. niloticus (Table 4). The intensity of infection in M. huberti was not significantly different between male (median: 5, range: 1–52) and female (median: 2, range: 1–39) rodents (U = 25.5; p = 0.37). Six *M. huberti* individuals (35.3%) harboured single male worms, whereas 11 (64.7%) presented at least one couple of paired male and female schistosomes. The overall parasite's sex ratio was in favour of males (102/89 = 1.15), including at the host species level in *M. huberti* (99/87 = 1.14) and *A. niloticus* (3/2 = 1.50). In total, 11 hosts (10 *M. huberti* and 1 A. niloticus) presented eggs in the liver and the mesenteries from which we hatched and collected live miracidia. Interestingly, one host (ID: JKIN_050) with a unique adult male Schistosoma showed signs of infection (granuloma to the liver). The successful hatching of miracidia proved the presence of an unseen female worm or an ancient infection. In two other hosts, which had worm pairs, no eggs or miracidia were found, indicating recent infections with sexually immature female worms. Details at the individual host level are available in S2 File.

Schistosomes were found in rodents during all sampling periods with variations in the overall prevalence and intensity (Table 4). The overall prevalence in *M. huberti* ranged from 2.4% in February-March 2022 to 9.3% in August 2022. In August 2022 and November 2021, when the majority of infected rodents were found, the intensity of infection could be substantial, reaching 23 and up to 52 worms in some individuals, a pattern not observed in other occasions Table 4. Prevalence and intensity of *Schistosoma* infection in rodents over time and across space. Only sites where infected hosts were detected are included in this table (the code in parentheses for each site refers to Fig 1). *M. huberti: Mastomys huberti, A. niloticus: Arvicanthis niloticus,* P: prevalence; I: intensity (median; range). "N-I" refers to the presence of only non-infected individuals (S2 File).

Zone-Village-Site	April 2021	November 2021	February-March 2022	August 2022	Overall periods
A-Temeye-Canal Sall (1a)	No data	<i>M. huberti</i> P: 1/3; I: 23	N-I	<i>M. huberti</i> P: 2/7; I: 27 (14; 1–26)	<i>M. huberti</i> P: 3/11; I: 50 (23; 1–26)
A-Temeye-Lewa (1b)	N-I	<i>M. huberti</i> P: 1/1; I: 1	N-I	<i>M. huberti</i> P: 1/1; I: 1	<i>M. huberti</i> P: 2/5; I: 2 (1)
A-Temeye-Lewa-Soubalo (1c)	No data	<i>M. huberti</i> P: 1/1; I: 5	N-I	<i>M. huberti</i> P: 1/7; I: 1	<i>M. huberti</i> P: 2/12; I: 6 (3; 1–5)
A-Temeye-Soubalo (1d)	N-I	<i>A. niloticus</i> P: 1/3; I: 5	<i>M. huberti</i> P: 2/6; I: 4 (2)	N-I	<i>M. huberti</i> P: 2/16; I: 4 (2)
					A. niloticus P: 1/12; I: 5
B-Guéo (4)	<i>M. huberti</i> P: 2/12; I: 7 (4; 1–6)	<i>M. huberti</i> P: 4/15; I: 103 (25; 2–52)	N-I	N-I	<i>M. huberti</i> P: 6/38; I: 110 (8; 1–52)
B-Gankette Guent (5)	<i>M. huberti</i> P: 1/25; I: 1	N-I	N-I	<i>M. hub<mark>erti</mark> P: 1/3; I: 13</i>	<i>M. huberti</i> P: 2/64; I: 14 (7; 1–13)
Overall zone A	N-I	<i>M. huberti</i> P: 3/19; I: 29 (5; 1–23)	<i>M. huberti</i> P: 2/22; I: 4 (2)	<i>M. huberti</i> P: 4/18; I: 29 (1; 1–26)	<i>M. huberti</i> P: 9/76; I: 62 (2; 1–26)
		<i>A. niloticus</i> P: 1/15; I: 5			<i>A. niloticus</i> P: 1/99; I: 5
Overall zone B	<i>M. huberti</i> P: 3/74; I: 8 (1; 1–6)	<i>M. huberti</i> P: 4/88; I: 103 (25; 2– 52)	N-I	<i>M. huberti</i> P: 1/36; I: 13	<i>M. huberti</i> P: 8/260; I: 124 (8; 1–52)
Overall zone A & B	<i>M. huberti</i> P: 3/91; I: 8 (1; 1–6)	<i>M. huberti</i> P: 7/107; I: 132 (10; 1–52)	<i>M. huberti</i> P: 2/84; I: 4 (2)	<i>M. huberti</i> P: 5/54; I: 42 (1; 1–26)	<i>M. huberti</i> P: 17/336; I: 186 (2; 1–52)
		A. niloticus P: 1/22; I: 5			A. niloticus P: 1/126; I: 5

https://doi.org/10.1371/journal.pntd.0012721.t004

(Table 4). Interestingly, August was also the time when rodents' capture rates and the proportion of juveniles in *M. huberti* were the lowest. However, probably due to a limited number of infected hosts and the high variance between hosts and over time, we did not find any significant difference in the overall prevalence (Fisher exact test: p = 0.24) and intensity (KW = 3.34, p = 0.34) across sampling periods in *M. huberti* (Table 4).

At the spatial scale, schistosome distribution in rodents was highly focalised. Infections occurred in both geographic zones but only within particular localities (3 villages out of the 8 studied) and within particular sites (6 sites out of 20) sometimes separated from the nearest, "non-infected" one, by only a few hundreds of meters. Zone A harboured more sites with infected rodents (n = 4) than zone B (n = 2), and the overall prevalence in *M. huberti* was significantly higher there also (Fisher exact test: p = 0.005). In contrast, differences of infection intensity between zones were not significant (U = 24.0, p = 0.26). At the scale of the villages, the prevalence in *M. huberti* significantly varied between 3.1% in Gankette Guent, 15.8% in Guéo, and 20.5% in Temeye (Fisher exact test: p = 0.01) and reached up to 40% in the specific site of Temeye Lewa (Table 4). The intensity of infection was highly variable between hosts but did not differ significantly across space in *M. huberti* (village: KW = 1.96; p = 0.38), probably due to a relatively low sample size.

Snail intermediate hosts sampling and Schistosoma infection

A total of 599 intermediate host snails were collected throughout our study (Table 5). Overall, we identified 318 *Biomphalaria* sp. (*Bi. pfeifferi*), and 281 *Bulinus* sp. (159 *Bu. truncatus*, 62 *Bu. globosus*, 43 *Bu. senegalensis* and 17 *Bu. forskalii*) according to shell morphology. The

Table 5. *Schistosoma* prevalence in intermediate host snail species sampled through time and space. The codes in parentheses for each site refers to Fig 1). Positive snails (shedding schistosome cercariae) are indicated with the symbol +. Other sites, not mentioned in this table, were not sampled for snails. Bi.: *Biomphalaria*, Bu.: *Bulinus*.

Zone-village-site	April 2021	November 2021	February-March 2022	August 2022
A-Teme-Canal sall (1a)	Not sampled	Bu. truncatus (0/1) Bu. globosus (0/16) Bu. senegalensis (0/6) Bu. forskalii (0/11)	Not sampled	Not sampled
A-Temeye-Lewa-Soubalo (1c)	Not sampled	Bu. globosus (0/1) Bu. forskalii (0/2)	Not sampled	Not sampled
A-Temeye-Soubalo (1d)	Bu. truncatus (0/21)	Bi. pfeifferi (0/21) Bu. truncatus (0/1)	Not sampled	Not sampled
A-Temeye-North (1e)	Bi. pfeifferi (0/27) Bu. globosus (0/5) Bu. forskalii (0/1)	Bu. globosus (0/9) Bu. senegalensis (0/5)	Not sampled	Bi. pfeifferi (0/10) Bu. truncatus (0/1) Bu. globosus (0/25) Bu. senegalensis (0/30) Bu. forskalii (0/2)
A-Mbane Center (2b)	Not sampled	Bi. pfeifferi (0/20) Bu. forskalii (0/1)	Not sampled	Not sampled
B-Gankette Balla (3)	Not sampled	Not sampled	Bi. pfeifferi (0/17) + Bu. truncatus (2/14) Bu. globosus (0/1)	+ Bi. pfeifferi (1/11) Bu. truncatus (0/2) Bu. senegalensis (0/1)
B-Guéo (4)	+ Bi. pfeifferi (7/37) Bu. truncatus (0/100)	Not sampled	+ Bi. pfeifferi (10/91) Bu. truncatus (0/16) Bu. globosus (0/3)	+ Bi. pfeifferi (1/6) Bu. truncatus (0/1)
B-Gankette Guent (5)	Bi. pfeifferi (0/24)	Bi. pfeifferi (0/7)	Bu. truncatus (0/1) Bu. globosus (0/1)	+ Bi. pfeifferi (6/47)
B-Merina Guewel-Takh (7c)	Not sampled	Not sampled	Not sampled	+ Bu. truncatus (1/1) Bu. globosus (0/1) + Bu. senegalensis (1/1)

https://doi.org/10.1371/journal.pntd.0012721.t005

abundance, diversity and infectivity of snails were not compared over time and across space because of our non-standardized sampling. However, amongst all, 29 snails (4.8%) shed schistosome cercariae (Table 5). The majority of infection occurred in *Bi. pfeifferi* followed by *Bu. senegalensis* and *Bu. truncatus* with an overall prevalence of 7.9% (25/318), 2.3% (1/43) and 1.9% (3/159), respectively (Fisher exact test: p = 0.01). *Biomphalaria pfeifferi* were found infected at all sampling periods except in November 2021. Infected *Bulinus* spp. were found in February-March and August 2022 (*Bu. truncatus* during both sampling periods, *Bu. senegalensis* in August only).

As for rodents, schistosomes infection in snails was highly focal. No positive molluscs were found in zone A, where positive rodents were found. All positive molluscs were found in 4 transmission sites of zone B, positive rodents being found in two of them. Infected *Bi. pfeifferi* were repeatedly found in Guéo with a local prevalence reaching 18.9% in April 2021 and only once in Gankette Guinth and Gankette Balla with a local prevalence of 12.8% and 9.1% respectively, in August 2022 (Table 5). Infected *Bu. truncatus* were found in two transmission sites (Gankette Balla and Takh) and *Bu. senegalensis* in a unique site (Takh) with a local prevalence ranging between 14.3% and 100% (Table 5).

Molecular characterisation of schistosomes from rodents and snails

Amongst all adult schistosomes collected from rodents (n = 191) *cox1* RD-PCR revealed the presence of 188 specimens assigned to *S. mansoni* (98.4%), one female worm to *S. bovis* and one male worm to *S. haematobium*, while one remaining male worm could not be identified (S3 File). All randomly selected *cox1* PCR amplicons sequenced (n = 17) confirmed profiles

Polymorphic positions	29	51	89	216	251	387	874	925	Species	Nb. of sequences
Ref (AF531312)	Α	Т	G	Т	Т	С	С	Α	Schistosoma rodhaini	-
Ref (AY446082)	Α	Т	Α	С	Т	С	Т	Α	Schistosoma man <mark>son</mark> i	-
Variant_Sm1	Α	Т	Α	С	Т	С	Т	Α		219/316 (69.3%)
Variant_Sm 2	Α	Т	A	С	Т	C	Т	A/T		65/316 (20.6%)
Variant_Sm 3	Α	C	A	С	Т	C	Т	Α		14/316 (4.4%)
Variant_Sm 4	Α	Т	Α	С	Т	С	Т	Т		7/316 (2.2%)
Variant_Sm 5	Α	Т	Α	С	С	С	Т	Α		3/316 (0.9%)
Variant_Sm 6	G	Т	A	С	Т	C	Т	Α		2/316 (0.6%)
Variant_Sm 7	Α	С	Α	С	Т	С	Т	A/T		2/316 (0.6%)
Variant_Sm 8	Α	Т	Α	С	Т	Т	Т	Т		1/316 (0.3%)
Variant_Sm 9	Т	С	Α	С	Т	С	Т	A		1/316 (0.3%)
Variant_Sm 10	A	Т	A	С	С	С	Т	A/T		1/316 (0.3%)
Variant_Sm 11	Α	Т	A	С	Т	Т	Т	A/T		1/316 (0.3%)

https://doi.org/10.1371/journal.pntd.0012721.t006

obtain from the RD-PCR except one unreadable sequence (S3 File). Sequencing of the ITS gene gave reliable data for 181 sequences while 10 sequences (all assigned to *S. mansoni cox1*) could not be interpreted due to bad quality or contaminations (S3 File). However, all individuals previously assigned to a species based on *cox1* had a congruent ITS sequence and none of the specimens examined showed polymorphism within the ITS sequence suggestive of hybridization (Tables 6 and 7).

A total of 69 miracidia collected from rodents (n = 11) were subject to *cox1* RD-PCR which confirmed the presence of *S. mansoni* in all of those infected individuals except one individual (where an unidentified adult male was previously found) with miracidia assigned to *S. haema-tobium* (n = 3 miracidia). These results were further confirmed by *cox1* sequencing (n = 10 miracidia). Sequencing of the ITS gene gave reliable results for 49 miracidia while 17 sequences could not be interpreted and 3 were not sequenced. No hybrid profile was found amongst the miracidia analysed.

These findings show the predominance of *S. mansoni* in rodents (16/18) through space and time but also the presence of *S. haematobium* in two *M. huberti* from Temeye Lewa. As shown from the presence of hatched miracidia, *S. haematobium*, which is usually restricted to humans, seems viable in rodents. Another interesting discovery was the presence of a *S. bovis* female paired with a male *S. mansoni* in one *M. huberti* from Temeye Soubalo. Unfortunately, as no eggs or miracida could be collected we could not evidence the presence of potential first-generation hybrids between these two species.

Polymorphic positions	70	380	553	722	777	827	897	930	Species	Nb. of sequences
Ref (MT884914)	A	A	С	G	С	G	С	А	Schistosoma haematobium	-
Ref (MT580950)	G	А	С	Α	Т	A	Т	А	Schistosoma bovis	-
Ref (OX103898)	G	А	С	A	Т	A	Т	Α	Schistosoma guineensis	-
Ref (MT580947)	Α	A	С	Α	Т	A	Т	А	Schistosoma curassoni	-
Variant_Sb1	G	A	С	Α	Т	A	Т	А	Schistosoma bovis	6/11 (54.5%)
Variant_Sb2	G	A	С	A	Т	A	Т	A/T		5/11 (45.5%)
Variant_Sh1	A	A	C/T	G	С	G	С	А	Schistosoma haematobium	5/8 (62.5%)
Variant_Sh2	A	A	C/T	G	С	G	С	A/T		2/8 (25.0%)
Variant Sh3	A	A/G	C/T	G	С	G	С	A/T		1/8 (12.5%)

Table 7. Polymorphic positions identified within the ITS gene (932 bp) between S. haematobium and S. bovis. Ref = reference sequence with GenBank accession.

https://doi.org/10.1371/journal.pntd.0012721.t007

Finally, over the 29 snails shedding schistosomes, 123 cercariae were analysed by cox1 RD-PCR. All cercariae (n = 103) emitted from *Bi. pfeifferi* (n = 25) were identified as *S. mansoni*, whereas cercariae (n = 18) shed from *Bu. truncatus* (n = 3) and *Bu. senegalensis* (n = 1) were either assigned to *S. bovis* (n = 16) or *S. haematobium* (n = 2). One of the *Bu. truncatus* was found infected with both *S. bovis* (n = 2 cercariae) and *S. haematobium* (n = 2 cercariae). The sequencing of 18 cox1 amplicons confirmed profiles obtained from the RD-PCR. Sequencing of the ITS gene gave reliable results for 105 cercariae while 14 could not be interpreted and 4 were not sequenced. All *S. mansoni cox1* profiles were congruent with the ITS sequences, thus no hybrids with this species were detected. However, although cercariae from two *Bu. truncatus* and one *Bu. senegalensis* had congruent results between cox1 and ITS profiles indicative of a *S. bovis* infection, one remaining *Bu. truncatus* was infected with both *S. haematobium* and *S. haematobium* x *S. bovis* hybrids (discordance between *S. bovis cox1* and *S. haematobium* ITS within the same individual). All molecular data at the individual parasite's level is available in <u>S3 File</u>.

Overall, we obtained 335 ITS gene sequences and 44 *cox1* sequences from schistosomes recovered form rodents and snails in the vicinity of Lake Guiers, Senegal. The majority of ITS sequences were assigned to *S. mansoni* (316 sequences: 179 adult worms, 47 miracidia and 90 cercariae) with 11 variants (Table 6). Double peaks were found in only one variant at position 925 and did not refer to *Schistosoma* species or known hybrids.

The remaining ITS sequences belong either to *S. bovis* (11 sequences with 2 variants) or *S. haematobium* (8 sequences with 3 variants). No double peaks were observed in the chromatograms at the five polymorphic positions (70, 722, 777, 827, and 897) between *S. haematobium*, *S. bovis* or *S. curassoni* (Table 7). However, double peaks were found at other unknown positions (380, 553, and 930) and did not refer to *Schistosoma* hybrids.

Discussion

The involvement of rodents in the transmission of schistosomiasis has raised concerns on the significant threat they may pose to the disease control efforts across Africa [12,20,26,27,29,31–33,53,58,84]. The role of vertebrate hosts other than humans has been largely overlooked in control strategies. Very little is known on the spatio-temporal variations of infection at the animal-parasite interface, a parameter that could help adapting the management of the disease. Based on a two-year survey around Lake Guiers, in northwestern Senegal, our study provides insight on the structure and population dynamics of small mammal communities as well as on their spatial and temporal variations of infection with schistosomes in this area.

The diversity of small mammal species was relatively low at transmission sites. Only four species were caught with overall capture rates were similar to those previously recorded in the area [26,27]. Very low capture rates were expected for *T. pygargus* living in dry habitats [63], however they were more surprising for *C. viaria* that is quite broadly distributed [85,86] including in humid habitats of the Senegal River Basin. As expected, the dominant species was *M. huberti*, known to inhabit humid and partially flooded environments in this regions [87]. Overall, small mammal capture rates varied between sampling periods, with lowest values observed during the rainy season. This observation, especially significant in *M. huberti* whose preferred habitat was systematically well sampled during all sampling periods, reflects lower rodent population sizes usually recorded in Sahelian environments at that period due to population turn over [28,63]. However, we cannot exclude that repeated sampling over time may have influenced small mammal abundance locally. For *A. niloticus*, the significant decline in capture rates after the first sampling period probably results from a lesser trapping effort after April 2021 along crop fields and gardens, which are the preferred habitats of this species [63],

as we focused on habitats close to water bodies to favour the capture of rodents infested by schistosomes. At the spatial scale, *M. huberti* was dominant in all localities except one (Mbane where various gardens were sampled) and there was a higher diversity of species in zone A, whilst only *M. huberti* and *A. niloticus* were trapped in zone B. A possible reason for this may be the higher diversity of habitats encountered in zone A, including humid riparian habitats boarding the lake, various crop fields and gardens, irrigation canals and livestock watering sites, compared to more homogeneous habitats along the lake in zone B. Concerning rodent population structure, there were no significant differences of sex ratio between sampling periods or geographical zones, within rodent species. On the contrary, the higher proportion of juveniles during the first semester (especially in February-March 2022), is congruent with the known pattern of reproduction of these rodents in Senegal [63,87,88]. The unusually high proportion of juveniles recorded for *A. niloticus* in August 2022 (zone A) may result from an exceptional reproductive event linked with local climatic or environmental conditions [87].

The prevalence of schistosomes in small mammal communities was relatively low on a global scale (3.6%), consistent with findings from other studies across Africa [26,27,29,31], in contrast to *foci* in the Caribbean [17,19], Southern America [18] or Asia [89,90]. However, Schistosoma infections seem to strongly depend on the spatio-temporal scale at which it is inferred from and on the host and parasites species considered [53]. As previously reported, both A. niloticus and M. huberti were found infected with schistosomes along Lake Guiers [26,27,29]. Mastomys huberti appeared as the main reservoir for schistosomes in this area as only one A. niloticus was found infected. Infected A. niloticus in Senegal were previously mainly found in the surroundings of Richard-Toll and Djidiery [26,27,29]. This observation and our results suggest local specificities driving the transmission of schistosomes in different rodent species. It is probable that differences in habitats found around Richard-Toll (i.e., large networks of crop fields and irrigation canals for the cultivation of rice and sugarcane, that are suitable for A. niloticus), and rodent species ecology, could at least in part, be at play. The use of different habitats and their proximity to various freshwater sources could explain the higher prevalence rates of schistosomes in *M. huberti* compared to *A. niloticus* occupying drier habitats in the Lake Guiers area (see also [63,87]). This therefore suggests the existence of contrasting epidemiological situations that may arise even at small spatio-temporal scales necessitating to assess Schistosoma infections in animals at a fine scale.

Infected rodents were found across all sampling periods, with no significant difference in prevalence or intensity of infection over time. However, most parasitized rodents and schistosomes were retrieved during or just after the rainy season, which coincides with the dominance of older individuals in rodent populations. Previous studies conducted in Richard-Toll also showed similar temporal variations in infection with a prevalence peak in July and November but no clear seasonal pattern over the years [28]. Nonetheless, the cumulative evidence of schistosomes found mostly, if not exclusively, in adult rodent individuals (this work, but also [28,29] and [26,27]) is not surprising. Indeed, rodents have a short lifespan and newborn individuals remain nested until two to three weeks of age [91]. Even if rodents rapidly come in contact with contaminated water sources, the development of mature adult worms that produce eggs and contribute to transmission takes several weeks (~5–7 weeks post-exposure). Therefore, it is unlikely to detect infected rodents less than two months old using current field methodologies (perfusion and dissections, coprology, miracidia hatching). Likewise, the rapid turnover in rodent populations may impose that the window of detection and transmission of the parasites from such hosts may be relatively short. The population dynamics and age structure of hosts thus appear as important parameters that can influence Schistosoma infections through time and space. Focusing on adult individuals of at least 3-4 months of age could be beneficial for future field studies conducted on rodents. Further research and the

development of new diagnostic methods and tools for animals easily employable in the field are needed in order to gain a deeper understanding of *Schistosoma* temporal transmission dynamics in small mammals, especially in relation to human reinfection after MDA of PZQ.

The distribution of infected rodents in the study area was highly focalised, a common feature of schistosomiasis transmission. Most sites where infected rodents were found and highest prevalence levels were observed along irrigation canals in zone A compared to sites with a direct access to the lake in zone B. This suggests that irrigation canals may be important sites for Schistosoma infections in both animals and human. Studies on schistosomiasis in human have indeed indicated that irrigation canals serve as hotspots for transmission despite MDA, likely due to factors such as proximity to households, ease in water accessibility, frequency of domestic and recreational activities and also a higher density of intermediate hosts snail populations [92]. It is important to note that although the prevalence in rodents could be high locally and temporally (reaching up to 100% in certain localities and at particular sampling periods), the hight variability of infection intensity at the individual host level highlights the dynamic role that rodents may play in schistosomes transmission. Therefore, both the prevalence and intensity of infection needs to be assessed to define the transmission potential of a particular host within a specific spatio-temporal context. Overall, our results indicate recurrent transmission of schistosomes in rodents at particular sites throughout several months (7–8 months) and more variation in Schistosoma infections across space than over time.

Recent records from villages along the Lac Guiers and the Senegal River basin show that in human populations S. mansoni is substantially prevalent (up to 45%) [27,45], but urogenital schistosomiasis is the most widespread (88% prevalence in children of which 72% where infected with S. haematobium-S. bovis hybrids) [45]. Likewise, the prevalence of schistosomes in sympatric livestock populations is extremely high (up to 94% for S. bovis in cattle). Interestingly, in rodents the most widespread schistosome species found was S. mansoni, confirming their role as true reservoir hosts in the global burden of intestinal schistosomiasis in that region [26,27,29]. Additionally, other schistosome species of human (S. haematobium) and animal (S. bovis) significance were sporadically found in rodent from Temeye. While S. bovis has been found in several wild rodents in sub-Saharan Africa [24,26,27,31,39,40], S. haematobium is considered as highly specific to humans [5,31,36]. The finding of two *M. huberti* infected with single male S. haematobium, including one with viable miracidia, suggests that M. huberti could be a spill-over host that may, to some extent, play a role in the transmission of urogenital schistosomiasis in Senegal. It also supports the need for more sensitive diagnostic tools for animal hosts, including detecting single sex infections that can easily pass unnoticed [93]. The presence of a heterospecific pair between a female S. bovis and a male S. mansoni in one M. huberti from Temeye was another notable finding. Rodents have been claimed as "biotic hubs" in which various *Schistosoma* species can meet and potentially hybridize [26,31,53]. For instance, Catalano et al. 2018 [26] previously reported a pair composed of a S. mansoni male with a hybrid female (ITS: S. haematobium x COI: S. bovis) in M. huberti from the same village, indicating that Temeye, may be a significant location for interactions between schistosome species. In our particular case, the female of the S. bovis x S. mansoni pair was immature (visual observation) and since the rodent infection was likely recent, no miracidia could be collected to confirm potential hybrids. Although these hybrids do not occur naturally, laboratory experiments have shown that S. mansoni and S. bovis can pair, producing parthenogenetic offspring that infect the maternal species' intermediate snail hosts [94]. This indicates that co-infection and heterospecific pairs in reservoir hosts like rodents, while not leading to hybridization per se, may still enable maternal species transmission without conspecific mates. The role of rodents as biotic hubs requires further clarification, as it has strong implications for the

parasites transmission and control [26,38,59]. In the same vein, although S. haematobium's capacity to infect rodents could have been due to genetic introgression from S. bovis, no such hybrids were found in rodents. The low prevalence of S. bovis and S. haematobium and the absence of hybrids in the rodent populations suggest behavioural divergence in local host-parasite interactions. However, this also highlights diverse epidemiological situations, where different animal reservoirs could locally play a role in parasite transmission. For instance, in Benin, M. natalensis and R. rattus, two rodent species absent in Northern Senegal, were found infected with S. bovis x S. haematobium hybrids [53]. This difference in local hosts-parasites interactions may be influenced by the cercarial emergence rhythm from the intermediate snail hosts, known as chronobiological polymorphism [95]. Indeed, S. bovis and S. haematobium typically follow a strictly diurnal rhythm, with cercarial emission peaking in the early morning and mid-day, respectively. However, S. bovis x S. haematobium hybrids in Benin exhibit atypical chronobiological patterns (early diurnal, late diurnal, nocturnal, double-peak emission patterns) allowing them to exploit a diversity of local hosts, including cattle, rodents and humans [49,53]. In Senegal, the chronobiological polymorphism of Schistosoma species, including hybrids needs to be assessed, especially since the presence of S. mansoni in both humans and the mainly nocturnal rodent M. huberti [96] could indicate the existence of various chronotypes [95].

Finally, we found Bi. pfeifferi, Bu. truncatus, and Bu. senegalensis shedding schistosomes near small mammals capture sites. Most infections occurred in Bi. pfeifferi, all transmitting S. mansoni, predominantly when snail populations peak in spring and early summer [97]. This is potentially congruent with a subsequent higher number of S. mansoni infected rodents in August and November, although our non-standardized opportunistic sampling did not allow us to compare snail infection levels through space and time. Infected Bulinus sp. were rarer with *Bu. truncatus* transmitting either *S. bovis* or *S. haematobium* and *S. haematobium x S.* bovis hybrids. We also provide further confirmation for Bu. senegalensis transmitting S. bovis in the area [98]. Positive snails were found in only 4 sites of zone B, overlapping with positive rodents in 2 sites. No positive molluscs were recovered in zone A, where positive rodents were caught. The inconsistency between snail and rodent infection can be attributed to several factors. First, the focality of transmission, along with the typically low prevalence of snail populations and our opportunistic sampling, likely led to missing infected snails. Second, the detection of infected rodents suggests that snails were shedding schistosomes several weeks earlier, meaning a temporal gap between hosts that could contribute to the observed patterns. Finally, the transmission at particular site of S. bovis and S. haematobium that seem to be rarer in rodents compared to S. mansoni, could further explain the absence of infected rodents in certain sites. Therefore, the absence of infected snails does not necessarily rule out ongoing transmission, while finding infected snails confirms active schistosomiasis transmission even if rodents are not infected. Assessing infections in definitive and intermediate hosts may provide complementary insights into schistosomiasis spatial and temporal occurrence.

In conclusion, our study emphasizes the role of rodents in the transmission of African schistosomes by acting as reservoirs for *S. mansoni* or spill-over hosts for *S. bovis* and *S. haematobium*. It highlights the fine spatio-temporal variations of infections in small mammal communities found in the vicinity of Lake Guiers in northern Senegal. Overall, infections were found predominantly during or just after the rainy season, a time when older individuals dominate in rodent populations. Variations in *Schistosoma* infections in rodents seemed to be greater over space than time, indicating the need for a better understanding of the parasite's epidemiology at various transmission sites. Our work also emphasises the role of rodents as biotic hubs allowing various *Schistosoma* species to encounter and potentially hybridize. The focality of transmission, the high variability in infection parameters of

rodents and molluscs that sometimes appear incongruent, and the relatively low number of infected rodents caught, make it difficult to drawn general patterns usable to adapt control strategies. New methods and tools enabling an easier monitoring of *Schistosoma* infection in animals would enable to conduct larger spatio-temporal surveys. Combined with fine scale population genomic tools, they should provide unprecedented resolution on schisto-somes zoonotic transmission, parasite diversity and their evolutionary relationships within and between humans, animals and the environment, allowing adapting local measures for a sustained control of the disease.

We would like to thank the various people that helped during fieldwork, Amadou Bamba Diop for hosting us in the health care centre of Mbane and village chiefs who welcomed us and helped locally to make these surveys possible.

Supporting information

S1 Table. Summary of the most parsimonious Generalized Linear Mixed Models (GLMMs) finally selected. AICc: Akaike's information criterion corrected for finite sample size. Δ : difference between the model selected and the model with the lowest AICc. LRT: Likelihood-ratio test.

(DOCX)

S1 File. Small mammal captures information over time and across space. (XLSX)

S2 File. Autopsy data from trapped small mammals over time and across space. (XLSX)

S3 File. Molecular data of adult worms, miracidia and cercariae found over time and across space in rodents and snails. (XLSX)

Author Contributions

Conceptualization: Julien Kincaid-Smith, Carine Brouat, Laurent Granjon.

- Formal analysis: Julien Kincaid-Smith, Boris Sègnito A. E. Savassi, Bruno Senghor, Christophe Diagne.
- Investigation: Julien Kincaid-Smith, Youssoupha Niang, Mamadou Kane, Caroline Tatard, Laurent Granjon.

Methodology: Julien Kincaid-Smith, Boris Sègnito A. E. Savassi, Bruno Senghor, Carine Brouat, Laurent Granjon.

Project administration: Julien Kincaid-Smith, Carine Brouat, Laurent Granjon.

Resources: Caroline Tatard.

Supervision: Carine Brouat, Laurent Granjon.

Visualization: Julien Kincaid-Smith.

Writing - original draft: Julien Kincaid-Smith, Carine Brouat, Laurent Granjon.

Writing – review & editing: Julien Kincaid-Smith, Boris Sègnito A. E. Savassi, Bruno Senghor, Christophe Diagne, Carine Brouat, Laurent Granjon.

References

- Ross AGP, Chau TN, Inobaya MT, Olveda RM, Li Y, Harn DA. A new global strategy for the elimination of schistosomiasis. International Journal of Infectious Diseases. 2017. pp. 130–137. https://doi.org/10. 1016/j.ijid.2016.09.023 PMID: 27939558
- 2. McManus DP, Dunne DW, Sacko M, Utzinger J, Vennervald BJ, Zhou XN. Schistosomiasis. Nat Rev Dis Prim. 2018; 4: 1–19. https://doi.org/10.1038/s41572-018-0013-8 PMID: 30093684
- 3. WHO. Guidelines on control and elimination of human schistosomiasis. Geneva: World Health Organization. 2022.
- 4. Boissier J, Mouahid G, Moné H. Schistosoma spp. Michigan State University; 2019.
- Rollinson D, Southgate VR, D R, V.R. S. The genus *Schistosoma*: a taxonomic appraisal. The biology of schistosomes From genes to latrines. 1987; 1–49. Available from: <u>http://www.cabdirect.org/</u> abstracts/19880851450.html
- 6. Webster JP, Molyneux DH, Hotez PJ, Fenwick A. The contribution of mass drug administration to global health: Past, present and future. Philos Trans R Soc B Biol Sci. 2014;369. https://doi.org/10.1098/rstb. 2013.0434 PMID: 24821920
- Zou HY, Yu QF, Qiu C, Webster JP, Lu DB. Meta-analyses of *Schistosoma japonicum* infections in wild rodents across china over time indicates a potential challenge to the 2030 elimination targets. PLoS Negl Trop Dis. 2020; 14: 1–21. https://doi.org/10.1371/journal.pntd.0008652 PMID: 32877407
- Boissier J, Moné H, Mitta G, Bargues MD, Molyneux D, Mas-Coma S. Schistosomiasis reaches Europe. Lancet Infect Dis. 2015; 15: 757–8. https://doi.org/10.1016/S1473-3099(15)00084-5 PMID: 26122434
- Moné H, Holtfreter MC, Allienne JF, Mintsa-Nguéma R, Ibikounlé M, Boissier J, et al. Introgressive hybridizations of *Schistosoma haematobium* by *Schistosoma bovis* at the origin of the first case report of schistosomiasis in Corsica (France, Europe). Parasitol Res. 2015; 114: 4127–4133. https://doi.org/ 10.1007/s00436-015-4643-4 PMID: 26268566
- WHO. Ending the neglect to attain the Sustainable Development Goals-A road map for neglected tropical diseases 2021–2030. Geneva: World Health Organization; 2020. Licence: CC BY-NC-SA 3.0 IGO.
- Gower CM, Vince L, Webster JP. Should we be treating animal schistosomiasis in Africa? The need for a One Health economic evaluation of schistosomiasis control in people and their livestock. Trans R Soc Trop Med Hyg. 2017; 111: 244–247. https://doi.org/10.1093/trstmh/trx047 PMID: 29044372
- Webster JP, Gower CM, Knowles SCL, Molyneux DH, Fenton A. One health—an ecological and evolutionary framework for tackling Neglected Zoonotic Diseases. Evol Appl. 2016; 9: 313–333. <u>https://doi.org/10.1111/eva.12341</u>. PMID: 26834828
- Chen MG. Schistosoma japonicum and japonicum-like infections: epidemiology, clinical and pathological aspects. Hum schistosomiasis. 1993; 237–270.
- 14. Zhou YB, Liang S, Jiang QW. Factors impacting on progress towards elimination of transmission of schistosomiasis japonica in China. Parasites and Vectors. 2012; 5: 1. https://doi.org/10.1186/1756-3305-5-275 PMID: 23206326
- Rudge JW, Webster JP, Lu DB, Wang TP, Fang GR, Basáñez MG. Identifying host species driving transmission of schistosomiasis japonica, a multihost parasite system, in China. Proc Natl Acad Sci USA. 2013; 110: 11457–11462. https://doi.org/10.1073/pnas.1221509110 PMID: 23798418
- Gordon CA, Kurscheid J, Williams GM, Clements ACA, Li Y, Zhou XN, et al. Asian schistosomiasis: Current status and prospects for control leading to elimination. Trop Med Infect Dis. 2019;4. https://doi. org/10.3390/tropicalmed4010040 PMID: 30813615
- 17. Théron A, Sire C, Rognon A, Prugnolle F, Durand P. Molecular ecology of *Schistosoma mansoni* transmission inferred from the genetic composition of larval and adult infrapopulations within intermediate and definitive hosts. Parasitology. 2004; 129: 571.
- Gentile R, Soares MS, Barreto M, Lessa Goncalves MM, D'Andrea PS et al. The Role of Wild Rodents in the Transmission of *Schistosoma mansoni* in Brazil. In: Rokni MB, editor. Schistosomiasis. IntechOpen; 2012. https://doi.org/10.5772/25909
- Théron A, Pointier JP, Morand S, Imbert-Establet D, Borel G. Long-term dynamics of natural populations of *Schistosoma mansoni* among *Rattus rattus* in patchy environment. Parasitology. 1992; 104: 291–298. https://doi.org/10.1017/S0031182000061734 PMID: 1594292
- Miranda GS, Rodrigues JGM, Silva JKA de O, Camelo GMA, Silva-Souza N, Neves RH, et al. New challenges for the control of human schistosomiasis: The possible impact of wild rodents in *Schistosoma mansoni* transmission. Acta Trop. 2022; 236: 106677. <u>https://doi.org/10.1016/j.actatropica.2022</u>. 106677 PMID: 36063905

- Panzner U, Boissier J. Natural intra-and interclade human hybrid schistosomes in africa with considerations on prevention through vaccination. Microorganisms. 2021; 9: 1–19. https://doi.org/10.3390/ microorganisms9071465 PMID: 34361901
- Kebede T, Bech N, Allienne JF, Olivier R, Erko B, Boissier J. Genetic evidence for the role of nonhuman primates as reservoir hosts for human schistosomiasis. PLoS Negl Trop Dis. 2020; 14: 1–20. https://doi.org/10.1371/journal.pntd.0008538 PMID: 32898147
- Richards L, Erko B, Ponpetch K, Ryan SJ, Liang S. Assessing the nonhuman primate reservoir of Schistosoma mansoni in Africa: A systematic review. Infect Dis Poverty. 2019; 8: 1–11. https://doi.org/10. 1186/s40249-019-0543-7 PMID: 31077256
- Standley CJ, Mugisha L, Dobson AP, Stothard JR. Zoonotic schistosomiasis in non-human primates: Past, present and future activities at the human-wildlife interface in Africa. J Helminthol. 2012; 86: 131– 140. https://doi.org/10.1017/S0022149X12000028 PMID: 22269859
- Müller-Graf CDM, Collins DA, Packer C, Woolhouse MEJ. Schistosoma mansoni infection in a natural population olive baboons (Papio cynocephalus anubis) in Gombe Stream National Park, Tanzania. Parasitology. 1997; 115: 621–627. https://doi.org/10.1017/S0031182097001698 PMID: 9488873
- Catalano S, Sène M, Diouf ND, Fall CB, Borlase A, Léger E, et al. Rodents as natural hosts of zoonotic Schistosoma species and hybrids: An epidemiological and evolutionary perspective from West Africa. J Infect Dis. 2018; 218: 429–433. https://doi.org/10.1093/infdis/jiy029 PMID: 29365139
- Catalano S, Léger E, Fall CB, Borlase A, Diop SD, Berger D, et al. Multihost Transmission of *Schistosoma mansoni* in Senegal, 2015–2018. Emerg Infect Dis J. 2020; 26: 1234. https://doi.org/10.3201/ eid2606.200107 PMID: 32441625
- Sène M, Godeluck B, Trape J-F, Duplantier J. Implications des rongeurs dans les problèmes de santé: l'exemple de la borréliose et de la bilharziose intestinale à Richard-Toll. In Hervé Jean-Pierre (ed.), Brengues Jacques (ed.). Aménagements hydro-agricoles et santé (vallée du fleuve Sénégal). Paris: ORSTOM, p. 185–199. (Colloques et Séminaires). Eau et Santé: Colloque, Dakar (SEN), 1994/11/14-15. ISBN 2-7099-1388-7. ISSN 0767-2896.
- **29.** Duplantier JM, Sene M. Rodents as reservoir hosts in the transmission of *Schistosoma mansoni* in Richard-Toll, Senegal, West Africa. J Helminthol. 2000; 74: 129–135.
- Sène M, Brémond P, Hervé JP, Southgate VR, Sellin B, Marchand B, et al. Comparison of human and murine isolates of *Schistosoma mansoni* from Richard-Toll, Senegal, by isoelectric focusing. J Helminthol. 1997; 71: 175–181. https://doi.org/10.1017/s0022149x00015868 PMID: 9192707
- Hanelt B, Mwangi IN, Kinuthia JM, Maina GM, Agola LE, Mutuku MW, et al. Schistosomes of small mammals from the Lake Victoria Basin, Kenya: new species, familiar species, and implications for schistosomiasis control. Parasitology. 2010; 137: 1109. https://doi.org/10.1017/S0031182010000041 PMID: 20380765
- **32.** Alehegne KD, Mitiku BA. *Schistosoma mansoni* Epidemiology Among Snails, Rodents and Children: A One Health Approach. Infect Drug Resist. 2022; 15: 5629–5643. https://doi.org/10.2147/IDR.S363953 PMID: 36187732
- Steinauer ML, Hanelt B, Mwangi IN, Maina GM, Agola LE, Kinuthia JM, et al. Introgressive hybridization of human and rodent schistosome parasites in western Kenya. Mol Ecol. 2008; 17: 5062–5074. <u>https:// doi.org/10.1111/j.1365-294X.2008.03957.x PMID: 18992007</u>
- Webster BL, Southgate VR, Timothy D, Littlewood J. A revision of the interrelationships of Schistosoma including the recently described *Schistosoma guineensis*. Int J Parasitol. 2006; 36: 947–955. https://doi. org/10.1016/j.ijpara.2006.03.005 PMID: 16730013
- Standley CJ, Dobson AP, Stothard JR, JC PA, Russell J. Out of animals and back again: schistosomiasis as a zoonosis in Africa. Schistosomiasis. InTech. 2012; 209–230. https://doi.org/10.5772/25567
- Pitchford RJ, Visser PS. The role of naturally infected wild rodents in the epidemiology of schistosomiasis in the Eastern Transvaal. Trans R Soc Trop Med Hyg. 1962; 56: 126–135. https://doi.org/10.1016/ 0035-9203(62)90139-6 PMID: 14486965
- Ouma JH, Fenwick A. Animal reservoirs of schistosomiasis. Parasit helminths zoonoses Africa. 1991; 224–236.
- Leger E, Webster JP. Hybridizations within the Genus Schistosoma: Implications for evolution, epidemiology and control. Parasitology. 2017. pp. 65–80. https://doi.org/10.1017/S0031182016001190 PMID: 27572906
- **39.** Ghandour AM, Babiker A. The in vivo development of *Schistosoma bovis* in the Nile rat. Ann Trop Med Parasitol. 1978; 72: 227–230.
- Lo CT, Lemma A. Studies on *Schistosoma bovis* in Ethiopia. Ann Trop Med Parasitol. 1975; 69: 375– 382. https://doi.org/10.1080/00034983.1975.11687022

- Teukeng FFD, Blin M, Bech N, Gomez MR, Zein-Eddine R, Simo AMK, et al. Hybridization increases genetic diversity in *Schistosoma haematobium* populations infecting humans in Cameroon. Infect Dis Poverty. 2022; 11: 37. https://doi.org/10.1186/s40249-022-00958-0 PMID: 35346375
- Angora EK, Allienne J-F, Rey O, Menan H, Touré AO, Coulibaly JT, et al. High prevalence of Schistosoma haematobium × Schistosoma bovis hybrids in schoolchildren in Côte d'Ivoire. Parasitology. 2019/ 11/21. 2020; 147: 287–294. https://doi.org/10.1017/S0031182019001549 PMID: 31727202
- Huyse T, Van den Broeck F, Hellemans B, Volckaert FAM, Polman K. Hybridisation between the two major African schistosome species of humans. Int J Parasitol. 2013; 43: 687–689. https://doi.org/10. 1016/j.ijpara.2013.04.001 PMID: 23643461
- Léger E, Borlase A, Fall CB, Diouf ND, Diop SD, Yasenev L, et al. Prevalence and distribution of schistosomiasis in human, livestock, and snail populations in northern Senegal: a One Health epidemiological study of a multi-host system. Lancet Planet Health. 2020; 4: e330–e342. https://doi.org/10.1016/ S2542-5196(20)30129-7 PMID: 32800151
- 45. Webster BL, Diaw OT, Seye MM, Webster JP, Rollinson D. Introgressive Hybridization of *Schistosoma* haematobium Group Species in Senegal: Species Barrier Break Down between Ruminant and Human Schistosomes. PLoS Negl Trop Dis. 2013; 7. https://doi.org/10.1371/journal.pntd.0002110 PMID: 23593513
- Boon NAM, Van Den Broeck F, Faye D, Volckaert FAM, Mboup S, Polman K, et al. Barcoding hybrids: heterogeneous distribution of *Schistosoma haematobium* × *Schistosoma bovis* hybrids across the Senegal River Basin. Parasitology. 2018: 1–12. https://doi.org/10.1017/S0031182018000525 PMID: 29667570
- Brémond P, Sellin B, Sellin E, Naméoua B, Labbo R, Théron A, et al. Arguments for the modification of the genome (introgression) of the human parasite *Schistosoma haematobium* by genes from *S. bovis*, in Niger. Comptes Rendus de L'academie des sciences. Serie III. Sciences de la vie, 1993; 316: 667– 670.
- Savassi BAES, Mouahid G, Lasica C, Mahaman SDK, Garcia A, Courtin D, et al. Cattle as natural host for *Schistosoma haematobium* (Bilharz, 1852) Weinland, 1858 x *Schistosoma bovis* Sonsino, 1876 interactions, with new cercarial emergence and genetic patterns. Parasitol Res. 2020; 119: 2189–2205. https://doi.org/10.1007/s00436-020-06709-0 PMID: 32468189
- Onyekwere AM, Rey O, Allienne JF, Nwanchor MC, Alo M, Uwa C, et al. Population Genetic Structure and Hybridization of *Schistosoma haematobium* in Nigeria. Pathogens. 2022; 11: 1–15. https://doi.org/ 10.3390/pathogens11040425 PMID: 35456103
- Privat A, Jérôme B, Bakary S, Laurent D, Assitan D, Niaré DS, et al. Genetic profiles of *Schistosoma haematobium* parasites from Malian transmission hotspot areas. Parasites and Vectors. 2023; 16: 1–8. https://doi.org/10.1186/s13071-023-05860-8 PMID: 37542265
- Boissier J, Grech-Angelini S, Webster BL, Allienne J-F, Huyse T, Mas-Coma S, et al. Outbreak of urogenital schistosomiasis in Corsica (France): an epidemiological case study. Lancet Infect Dis. 2016; 16: 1–9. https://doi.org/10.1016/S1473-3099(16)00175-4 PMID: 27197551
- Ramalli L, Mulero S, Noël H, Chiappini JD, Vincent J, Barré-Cardi H, et al. Persistence of schistosomal transmission linked to the cavu river in southern Corsica since 2013. Eurosurveillance. 2018; 23: 2–5. https://doi.org/10.2807/1560-7917.ES.2018.23.4.18–00017
- 53. Savassi BAES, Dobigny G, Etougbétché JR, Avocegan TT, Quinsou FT, Gauthier P, et al. Mastomys natalensis (Smith, 1834) as a natural host for *Schistosoma haematobium* (Bilharz, 1852) Weinland, 1858 x Schistosoma bovis Sonsino, 1876 introgressive hybrids. Parasitol Res. 2021; 120: 1755–1770. https://doi.org/10.1007/s00436-021-07099-7 PMID: 33687566
- Oleaga A, Rey O, Polack B, Grech-Angelini S, Quilichini Y, Pérez-Sánchez R, et al. Epidemiological surveillance of schistosomiasis outbreak in Corsica (France): Are animal reservoir hosts implicated in local transmission? PLoS Negl Trop Dis. 2019; 13. https://doi.org/10.1371/journal.pntd.0007543 PMID: 31233502
- 55. Taylor M. Hybridization experiments on five species of African schistosomes. J Helminthol. 1970; 44: 253–314.
- Han BA, Kramer AM, Drake JM. Global Patterns of Zoonotic Disease in Mammals. Trends Parasitol. 2016; 32: 565–577. https://doi.org/10.1016/j.pt.2016.04.007 PMID: 27316904
- Han BA, Schmidt JP, Bowden SE, Drake JM. Rodent reservoirs of future zoonotic diseases. Proc Natl Acad Sci USA. 2015; 112: 7039–7044. https://doi.org/10.1073/pnas.1501598112 PMID: 26038558
- Streicker DG, Fenton A, Pedersen AB. Differential sources of host species heterogeneity influence the transmission and control of multihost parasites. Ecol Lett. 2013; 16: 975–984. https://doi.org/10.1111/ ele.12122 PMID: 23714379

- Colley DG, Loker ES. New tools for old questions: How strictly human are "Human Schistosomes"— And Does It Matter? J Infect Dis. 2018; 218: 344–346. https://doi.org/10.1093/infdis/jiy030 PMID: 29365121
- 60. Knowles SCL, Webster BL, Garba A, Sacko M, Diaw OT, Fenwick A, et al. Epidemiological Interactions between Urogenital and Intestinal Human Schistosomiasis in the Context of Praziguantel Treatment across Three West African Countries. PLoS Negl Trop Dis. 2015; 9: 1–25. https://doi.org/10.1371/ journal.pntd.0004019 PMID: 26469347
- Lund AJ, Sam MM, Sy AB, Sow OW, Ali S, Sokolow SH, et al. Unavoidable risks: Local perspectives on water contact behavior and implications for schistosomiasis control in an agricultural region of Northern Senegal. Am J Trop Med Hyg. 2019; 101: 837–847. https://doi.org/10.4269/ajtmh.19-0099 PMID: 31452497
- Sikes R. S., Animal Care and Use Committee of the American Society of Mammalogists. 2016 Guidelines of the American Society of Mammalogists for the use of wild mammals in research and education. Journal of mammalogy, 2016; 97(3), 663–688. https://doi.org/10.1093/jmammal/gyw078
- Granjon L, Duplantier J-M. Les rongeurs de l'Afrique sahélo-soudanienne. Editions de l'IRD, Marseille, France, 2009. Available from: https://www.documentation.ird.fr/hor/fdi:010048662
- 64. Granjon L, Ba K, Diagne C, Ndiaye A, Piry S, Thiam M. La communauté des petits rongeurs du Ferlo: Tendances historiques et caractéristiques du peuplement actuel. In La Grande Muraille Verte, une Réponse Africaine au Changement Climatique. 2019; 1–19.
- Dumont M, Moné H, Mouahid G, Idris MA, Shaban M, Boissier J. Influence of pattern of exposure, parasite genetic diversity and sex on the degree of protection against reinfection with Schistosoma mansoni. Parasitol Res. 2007; 101: 247. https://doi.org/10.1007/s00436-007-0476-0 PMID: 17310396
- 66. Yu JM, de Vlas SJ, Jiang QW, Gryseels B. Comparison of the Kato-Katz technique, hatching test and indirect hemagglutination assay (IHA) for the diagnosis of Schistosoma japonicum infection in China. Parasitol Int. 2007; 56: 45–49. https://doi.org/10.1016/j.parint.2006.11.002 PMID: 17188018
- Allan F, Dunn AM, Emery AM, Stothard JR, Johnston DA, Kane RA, et al. Use of sentinel snails for the detection of *Schistosoma haematobium* transmission on Zanzibar and observations on transmission patterns. Acta Trop. 2013; 128: 234–240. https://doi.org/10.1016/j.actatropica.2013.01.003 PMID: 23318933
- Pennance T, Ame SM, Amour AK, Suleiman KR, Allan F, Rollinson D, et al. Occurrence of Schistosoma bovis on Pemba Island, Zanzibar: implications for urogenital schistosomiasis transmission monitoring. Parasitology. 2018; 1–5. https://doi.org/10.1017/S0031182018001154 PMID: 30086805
- Jones IJ, Sokolow SH, Chamberlin AJ, Lund AJ, Jouanard N, Bandagny L, et al. Schistosome infection in senegal is associated with different spatial extents of risk and ecological drivers for *Schistosoma haematobium* and *S. mansoni*. PLoS Negl Trop Dis. 2021; 15: 1–24. https://doi.org/10.1371/journal.pntd. 0009712 PMID: 34570777
- Brown DS. Freshwater Snails Of Africa And Their Medical Importance. 2nd Ed. London, UK: Taylor and Francis (1994). https://doi.org/10.1201/9781482295184
- Webster BL, Rabone M, Pennance T, Emery AM, Allan F, Gouvras A, et al. Development of novel multiplex microsatellite polymerase chain reactions to enable high-throughput population genetic studies of Schistosoma haematobium. Parasites and Vectors. 2015; 8: 1–5. <u>https://doi.org/10.1186/s13071-015-1044-6 PMID: 26329827</u>
- 72. Van den Broeck F, Geldof S, Polman K, Volckaert FAM, Huyse T. Optimal sample storage and extraction procotols for reliable multilocus genotyping of the human parasite Schistosoma mansoni. Infect Genet Evol. 2011; 11: 1413–1418. https://doi.org/10.1016/j.meegid.2011.05.006 PMID: 21605705
- Webster BL, Rollinson D, Stothard JR, Huyse T. Rapid diagnostic multiplex PCR (RD-PCR) to discriminate Schistosoma haematobium and S. bovis. J Helminthol. 2010; 84: 107–114. https://doi.org/10. 1017/S0022149X09990447 PMID: 19646307
- Kane RA, Rollinson D. Repetitive sequences in the ribosomal DNA internal transcribed spacer of *Schistosoma haematobium*, *Schistosoma intercalatum* and *Schistosoma mattheei*. Mol Biochem Parasitol. 1994; 63: 153–156. https://doi.org/10.1016/0166-6851(94)90018-3 PMID: 8183315
- 75. Edgar RC. MUSCLE: Multiple sequence alignment with high accuracy and high throughput. Nucleic Acids Res. 2004; 32: 1792–1797. https://doi.org/10.1093/nar/gkh340 PMID: 15034147
- 76. Kumar S, Stecher G, Tamura K. MEGA7: Molecular Evolutionary Genetics Analysis Version 7.0 for Bigger Datasets. Mol Biol Evol. 2016; 33: 1870–1874. <u>https://doi.org/10.1093/molbev/msw054</u> PMID: 27004904
- 77. Huyse T, Webster BL, Geldof S, Stothard JR, Diaw OT, Polman K, et al. Bidirectional introgressive hybridization between a cattle and human schistosome species. PLoS Pathogens. 2009; 5(9): e1000571. https://doi.org/10.1371/journal.ppat.1000571 PMID: 19730700

- Johnson JB, Omland KS. Model selection in ecology and evolution. Trends Ecol Evol. 2004; 19: 101– 108. https://doi.org/10.1016/j.tree.2003.10.013 PMID: 16701236
- Arnold TW. Uninformative parameters and model selection using Akaike's Information Criterion. J Wildl Manage. 2010; 74: 1175–1178.
- **80.** Burnham KP, Anderson DR, Huyvaert KP. AIC model selection and multimodel inference in behavioral ecology: some background, observations, and comparisons. Behav Ecol Sociobiol. 2011; 65: 23–35.
- Hartig F. DHARMa: Residual Diagnostics for Hierarchical (Multi-Level/Mixed) Regression Models. R package version 0.3.3.0. (2020). Available online at: https://CRAN.R-project.org/package=DHARMa
- 82. Barton K, Barton MK. Package 'MuMIn' Multi-model inference, R package version 1436. https:// RForge.R-project.org/projects/mumin, 2019.
- De Boeck P, Bakker M, Zwitser R, Nivard M, Hofman A, Tuerlinckx F, et al. The estimation of item response models with the Imer function from the Ime4 package in R. J Stat Softw. 2011; 39: 1–28.
- 84. Hanelt B, Brant S V., Steinauer ML, Maina GM, Kinuthia JM, Agola LE, et al. Schistosoma kisumuensis n sp. (Digenea: Schistosomatidae) from murid rodents in the Lake Victoria Basin, Kenya and its phylogenetic position within the S. haematobium species group. Parasitology. 2009/07/01. 2009; 136: 987– 1001. https://doi.org/10.1017/S003118200900643X PMID: 19573258
- Hutterer R. Crocidura viaria.—African Savanna Path Shrew. In Happold M. & Happold D. C. D. (Eds.), Mammals of Africa 2013, (Vol. IV, pp. 142–143). Bloomsbury Publishing, London.
- Jacquet F, Denys C, Verheyen E, Bryja J, Hutterer R, Kerbis Peterhans JC, et al. Phylogeography and evolutionary history of the Crocidura olivieri complex (Mammalia, Soricomorpha): From a forest origin to broad ecological expansion across Africa. BMC Evol Biol. 2015; 15: 1–15. https://doi.org/10.1186/ s12862-015-0344-y PMID: 25900417
- Niang CT, Kane M, Niang Y, Sarr N, March L, Tatard C, et al. Socio-environmental changes and rodent populations in lowland agroecosystems of the lower delta of the River Senegal, West Africa: results of observations over a decade, 2008–2019. J Vertebr Biol. 2022; 71: 2008–2019. https://doi.org/10. 25225/jvb.22015
- 88. Poulet AR. The ecological basis of forecasting rodent outbreaks in a Sahelian agrosystem. Acta Zool Fenn. 1985; 173: 107–111.
- Lu DB, Wang TP, Rudge JW, Donnelly CA, Fang GR, Webster JP. Contrasting reservoirs for *Schisto-soma japonicum* between marshland and hilly regions in Anhui, China a two-year longitudinal parasitological survey. Parasitology. 2010; 137: 99–110. https://doi.org/10.1017/S003118200999103X PMID: 19723358
- 90. Fernandez TJ, Tarafder MR, Balolong E, Joseph L, Willingham AL, Bélisle P, et al. Prevalence of Schistosoma japonicum infection among animals in fifty villages of Samar Province, the Philippines. Vector-Borne Zoonotic Dis. 2007; 7: 147–155. https://doi.org/10.1089/vbz.2006.0565 PMID: 17627431
- **91.** Neal BR. Observations on the early post-natal growth and development of Tatera leucogaster, *Aeth-omys chrysophilus* and *A. namaquensis* from Zimbabwe, with a review of the pre- and post-natal growth and development of African Muroid rodents. Mammalia. 1990; 54: 245–270. <u>https://doi.org/10.1515/mamm-1990-0209</u>
- 92. Senghor B, Mathieu-Begné E, Rey O, Doucouré S, Sow D, Diop B, et al. Urogenital schistosomiasis in three different water access in the Senegal river basin: prevalence and monitoring praziquantel efficacy and re-infection levels. BMC Infect Dis. 2022; 22: 0–12. https://doi.org/10.1186/s12879-022-07813-5 PMID: 36581796
- Lu DB, Deng Y, Ding H, Liang YS, Webster JP. Single-sex schistosome infections of definitive hosts: Implications for epidemiology and disease control in a changing world. PLoS Pathog. 2018; 14: 1–7. https://doi.org/10.1371/journal.ppat.1006817 PMID: 29494686
- Onyekwere AM, De Elias-Escribano A, Kincaid-smith J, Dametto S, Allienne J-F, Rognon A, et al. Mating Interactions between *Schistosoma bovis* and *S. mansoni* and Compatibility of Their F1 Progeny with Biomphalaria glabrata and Bulinus truncatus. Microorganisms. 2022; 10: 1–10. https://doi.org/10.3390/ microorganisms10061251 PMID: 35744769
- 95. Théron A. Chronobiology of trematode cercarial emergence: From data recovery to epidemiological, ecological and evolutionary implications. Advances in Parasitology. Elsevier Ltd; 2015. https://doi.org/ 10.1016/bs.apar.2015.02.003 PMID: 25911367
- 96. Duplantier J-M, Granjon L. Rythmes d'activité chez six espèces de Muridés du Sénégal appartenant aux genres Mastomys, Arvicanthis, Myomys et Dasymys. Mammalia. 1990; 54: 173–182. Available: https://www.documentation.ird.fr/hor/fdi:35201
- 97. Sturrock RF, Diaw OT, Talla I, Niang M, Piau JP, Capron A. Seasonality in the transmission of schistosomiasis and in populations of its snail intermediate hosts in and around a sugar irrigation scheme at

Richard Toll, Senegal. Parasitology. 2001;123. <u>https://doi.org/10.1017/s0031182001008125</u> PMID: 11769294

 Senghor B, Webster B, Pennance T, Sène M, Doucouré S, Sow D, et al. Molecular characterization of schistosome cercariae and their *Bulinus* snail hosts from Niakhar, a seasonal transmission focus in central Senegal. Curr Res Parasitol Vector-Borne Dis. 2023; 3. https://doi.org/10.1016/j.crpvbd.2023. 100114 PMID: 36824299